RESEARCH ARTICLE



Treatment mechanism of the WHO caregiver skills training intervention for autism delivered in community settings

Michele Settanni¹ Chiara Servili⁴ D 1

| Katharine Suma² Erica Salomone^{4,5}

| Lauren B. Adamson D | Helen McConachie³

I

¹Department of Psychology, University of Turin, Turin, Italy

²Department of Human Development and Family Science, University of Georgia, Athens, Georgia, USA

³Population Health Sciences Institute, Newcastle University, Newcastle upon Tyne, UK

⁴Department of Mental Health and Substance Use, World Health Organization, Geneva, Switzerland

⁵Department of Psychology, University of Milano-Bicocca, Milan, Italy

Correspondence

Erica Salomone, Department of Psychology, University of Milano-Bicocca, Piazza dell'Ateneo Nuovo 1, 20126 Milano, Italy. Email: erica.salomone@unimib.it

Funding information

Autism Speaks; Compagnia di San Paolo, Bando Intrecci, Grant/Award Number: #2016. AAI4482 U5295: Train to Move (T2M) Grant/Award Number: 609402 - 2020; H2020 Marie Skłodowska-Curie Actions

Abstract

This study examined the mechanism of effect of the WHO Caregiver Skills Training (CST) through secondary analysis of a pilot RCT conducted in community settings. Participants were 86 caregivers (77% mothers) of children with ASD (78% male, mean age: 44.8 months) randomized to CST (n = 43) or treatment as usual (n = 43). The primary outcomes, measured at baseline (t1), immediately post-intervention (t2), and 3 months post-intervention (t3), were derived from the coding of caregiver-child free play interactions with the Brief Observation of Social-Communication Change (BOSCC) and the Joint Engagement Rating Inventory scale (JERI). At t3 positive treatment main effects had been observed for caregiver skills supportive of the interaction and for flow of the interaction (JERI), albeit only non-significant changes in the expected direction for child outcomes: autism phenotypic behaviors (BOSCC), joint engagement and availability to interact (JERI). This study tested the theory of change of CST, hypothesizing that the intervention would lead to an improvement on all child and dyad outcomes through an increase in the caregiver skills supportive of the interaction. Serial mediation analyses revealed that the effect of the intervention was significantly influenced by change in caregiver skills. Participation in the intervention led to notable increases in caregiver skills at t2 and t3, which subsequently contributed to improvements at t3 in flow of the interaction, autism phenotypic behavior, joint engagement, and availability to interact. We confirmed our a priori hypothesis showing that change in caregiver skills significantly mediated the effect of treatment on the dyad primary outcome, as well as on the other child outcomes that had shown non-significant changes in the expected direction. Implications for intervention design and policy making in the context of public health services are discussed.

[†] Deceased.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made. © 2023 The Authors. Autism Research published by International Society for Autism Research and Wiley Periodicals LLC.

Lay Summary

Community-implemented caregiver-mediated interventions are commonly used with children with autism spectrum disorder under age 5, but few studies have examined their effective mechanism (i.e., the processes responsible for changes in the clinical outcomes). This study examined the mechanism of a recently developed intervention, the WHO Caregiver Skills Training (CST), within a randomized controlled trial carried out in the community. The trial had shown positive intervention effects on parent skills supportive of the interaction and for fluent interaction between the child and caregiver, but less strong changes in the expected direction for child autism characteristic behaviors, joint engagement and availability to interact. The current study tested the hypothesis that CST works by increasing the parent's skills in support of the interaction, which in turn can lead to improved outcomes. We found that caregivers who received the CST intervention improved significantly more in their interaction skills with the child, and this led to a significant improvement on all joint interaction and child outcomes, including those which did not show a main effect of the intervention. This study shows that an acceptable and feasible open access community-implemented caregiver-mediated intervention can lead to a sustained improvement in the quality of the caregiverchild interaction through enhancing the caregiver's interaction skills.

KEYWORDS

caregiver-mediated intervention, mediation, parent training, parent-child interaction, RCT design, treatment mechanism, WHO Caregiver Skills Training, WHO CST

INTRODUCTION

Despite the tremendous progress in autism intervention research in the past decades, there remain wide unknowns about what kind of treatments should be provided "when, to whom, for how long, with what expected outcomes, and for what cost" (Lord et al., 2021, p. 272). Understanding the "why" and "how" a treatment is efficacious (or is not) is crucial to inform intervention design and, ultimately, in the context of pressure on public health systems globally, essential to support effective health policy decisions. Increasingly, research in child and adolescent mental health has addressed the question of "how" by performing formal tests of the hypothesized mechanism of interventions through mediation models. These allow examination of whether the expected outcomes have been brought forward by the pre-specified intervention active components, and potentially how the treatment effects may continue beyond the intervention period (Charman, 2021). Moreover, mediation studies of treatment mechanism are in themselves fundamentally relevant for developmental science and developmental social neuroscience since they can be conceptualized as analyses of how a given "developmental perturbation" (the intervention) affects the longitudinal course of complex conditions such as neurodevelopmental disorders through enhancing the child-environment fit (Green & Dunn 2008; Dawson, Rieder & Johnson 2023).

However, in the autism field, there have been few formal tests of treatment mechanism in caregiver- or teacher-mediated interventions, an intervention type very commonly used for children under 5 years

(Lord et al., 2021) and supported by evidence of changes in proximal outcomes particularly in the social communication domain (Crank et al., 2021). Pickles et al. (2015) showed that "parent synchrony" (proportion of parent communication acts that are a synchronous response to a child communication act) mediated the extent of autism phenotypic behaviors on a separate, examiner-led assessment (the Calibrated Severity Score, CSS, on the ADOS-2; Lord et al., 2012) in the Preschool Autism Communication Trial (PACT). In a trial of the Joint Attention, Symbolic Play, Engagement and Regulation (JASPER) intervention against psychoeducation, "mirrored pacing" (quality and frequency of parent imitation of child play acts) mediated duration of joint engagement in the same parent-child interaction (Gulsrud et al., 2016). Increased parental sense of competence mediated the reduction of child challenging behaviors not only at intervention endpoint but also at a later follow-up in a large cluster randomized trial of "An Individualized Mental Health Intervention for ASD" (AIM HI) training for behaviors that challenge (Brookman-Frazee et al., 2021). Furthermore, secondary analyses of RCTs of the JASPER (Shih et al., 2021) and PACT interventions (Carruthers et al., in press) demonstrated how sustained changes in child skills (joint engagement in JASPER, child initiations in PACT) in turn led to improvements in, respectively, expressive and receptive language on the Mullen Scales of Early Learning and the extent of autism behaviors on the ADOS-2. Taken together, this evidence points to promising pathways to enhancing child skills by improving parent-child interaction outcomes.

Nonetheless, it is important to consider that the effect sizes of child clinical outcomes in caregiver-mediated interventions implemented in community settings are, not surprisingly, considerably smaller than those reported in trials conducted in university-based settings (Nahmias et al., 2019). This reflects a number of challenges, including difficulties maintaining high levels of fidelity across both developmental strategies and behavioral strategies, which are less frequently embedded by early intervention providers in their practice (Pickard et al., 2021). In this perspective, the systematic search for "essential" intervention components can support the progressive optimization of effective treatments implemented in community settings (Williams & Beidas, 2019). Studies of treatment mechanisms of interventions delivered in community settings and formally evaluated for acceptability and feasibility are therefore particularly needed. The accurate reporting of acceptability and feasibility is, in fact, strategic in order to determine whether a lack of (or reduced) intervention effect may fundamentally be due to implementation issues rather than higher-order considerations that pertain to the theory of change underlying the intervention. However, of the few above mentioned autism intervention studies which conducted formal mediation analyses, only two report on trials of interventions implemented in community settings: a teacher-mediated model delivered in public preschools (Shih et al., 2021) and a caregiver-mediated model delivered to parents by community providers trained by trial therapists (Brookman-Frazee et al., 2021); the latter especially stands out from narrower efficacy trials for its complex cascade training model, as previously noted (Charman, 2021). Nonetheless, more evidence is needed on the mechanisms of effect of interventions conducted in the community.

The current study

This study examines the mechanisms of effect of a caregiver-mediated intervention implemented in the community, the "Caregiver Skills Training for families of children with developmental delays or disabilities" (CST). The CST, an open access program developed by WHO for developmental delays or disabilities, including autism (Salomone et al., 2019; WHO, 2022) is based on naturalistic developmental behavioral intervention methodologies (NDBI; Schreibman et al., 2015). CST comprises nine group sessions and three home visits training caregivers through adult-learning methods on the topics of engagement, play and home routines, communication, challenging behavior, daily living skills, parent wellbeing, and problem solving (Salomone et al., 2019); interventionist guides and participant booklets are freely available for download from the WHO website (https://www. who.int/teams/mental-health-and-substance-use/treatmen t-care/who-caregivers-skills-training-for-families-of-childr en-with-developmental-delays-and-disorders).

In a pilot RCT of CST provided in public health services in Italy with excellent acceptability to caregivers and feasibility of delivery (Salomone, Ferrante, et al., 2021), some relevant favorable treatment effects were reported on caregiver/child interaction data (Salomone, Settanni, et al., 2021). Specifically, at the 3-months follow-up on the blind-rated primary outcomes positive treatment effects were observed for parent skills supportive of the interaction and for dyadic fluency of the interaction, albeit not for child autism severity, joint engagement, and availability to interact; on these outcomes non-significant changes were observed in the expected direction.

As per registered protocol, the intervention theory of change posits that participation in the intervention is expected to lead to an improvement in the child's availability to interact, in joint engagement and flow of interaction through an increase in the parent's skills in support of the interaction. In this study, we therefore tested the treatment mechanism of CST examining if participation in the intervention would improve parent skills supportive of the interaction at treatment endpoint and at the 3-months follow-up, and if this, in turn, would mediate improvement on all dyad and child outcomes at follow-up, including those which had not shown main treatment effects. According to Hayes (2017), the presence of a significant indirect effect confirms mediation, even without a significant total effect: "A failure to test for indirect effects in the absence of a total effect can lead to you miss some potentially interesting, important, or useful mechanisms by which X exerts some kind of effect on Y" (Hayes, 2009, p. 415). With this study we therefore aimed to identify possible indirect treatment effects to shed light on the intervention mechanism and potentially to identify strategies to boost its effectiveness.

METHODS

Design

The pilot CST trial was a 6-center, 2-arm assessorblinded RCT of CST against enhanced treatment as usual (eTAU: one psychoeducation session in addition to TAU) of 86 children with autism aged 24-60 months. This study was performed in line with the principles of the Declaration of Helsinki. The trial was retrospectively registered (trial registry: ISRCTN, registration identification number: ISRCTN11702408, www.isrctn.com), as at the time of drafting of the study protocol the caregiver/ child interaction measure that was selected to measure the blind-rated primary outcomes (an adaptation of the JERI scale developed for WHO, Adamson et al., 2020) was not yet available. Ethical approval was granted by the Institutional Review Boards of the University of Turin (#57273) and of all National Health Service study sites: ASL Città di Torino (#0010244 and #0077761),

ASL AL (#AslAL.NPI.16.01), ASL CN1 (#180-16), ASL BI (#CE128/17). Both parents gave written consent. The primary outcome paper (Salomone, Settanni, et al., 2021) followed CONSORT guidelines. Families in

et al., 2021) followed CONSORT guidelines. Families in the CST intervention group attended 9 weekly group sessions and three home visits (before the first, after the 4th and after the 9th session) over 3 months. Data were collected at baseline (t1), immediately post-intervention (3 months post-baseline, t2), and 3 months postintervention (t3; Figure 1).

Participants

Participants were caregivers of children (n = 86) recruited as per the following inclusion criteria: (a) child age between 24 and 60 months; (b) child local clinical diagnosis of ASD by ICD-10 criteria (obtained with a combination of semi-structured observations, parent interviews and school reports) confirmed by research assessments. The exclusion criteria were: (a) level of spoken Italian in the caregiver not sufficient for a full participation in the intervention; (b) psychiatric conditions in the caregiver as reported in the clinical notes; these criteria were selected

to ensure that parents could fully participate in the intervention. Children were not excluded based on level of intellectual disability or any co-occurring condition. Participants were recruited from Child Neuropsychiatry Services of the national public health system in Italy; the sample consisted of mostly male (n = 67; 77.9%) preschoolers (age in months: M = 44.88; SD = 9.52) with moderate levels of autism phenotypic behaviors on the ADOS-2 CSS (M = 6.94; SD = 21.17) and developmental delay as measured on the Griffiths III (age equivalents scores for the General Development composite: M = 23.34; SD = 6.93). The two arms did not differ on any demographic or clinical characteristics. The intervention was delivered by local clinicians: one child neuropsyeight clinical psychologists chiatrist. and three "psychomotor therapists" (healthcare professionals operating in the neuro-psychomotor, neuropsychology and developmental psychopathology areas). The clinicians received a 5-day training course including presentations, role plays and practice with volunteer families and four post-training supervised practice sessions with families delivered by a WHO CST Team member (ES); all interventionists met post-training CST competency criteria. The interventionists demonstrated faithful and correct



FIGURE 1 CONSORT 2010 flow diagram (adapted from http://www.consort-statement.org/).

implementation both of the group sessions component (range of integrity ratings: 82%–97%), and of the home visits component (range of fidelity of use of intervention strategies in interaction with children: 73%–80%; Salomone, Settanni, et al., 2021). Access to treatment as usual was not restricted for either arm and the two groups did not differ on hours of child-directed services, school support, parenting programs or psychoeducational counseling received during the trial. In the treatment arm, participation in the program was open to 1–2 caregivers per family; for both groups data were collected on a target caregiver/child dyad designated by the family at baseline. For more details on the methodology of the trial, see previous reports (Salomone, Ferrante, et al., 2021; Salomone, Settanni, et al., 2021).

Outcomes

The child and dyad primary outcome measures were derived from the coding of 10 consecutive minutes of an approximately 12-min free play caregiver-child interaction videorecorded at the child's home (n = 256)interactions). Parents were given a standard toy kit suitable for a range of developmental play levels and instructed to play as they would usually do; coding began as soon as either partner had referenced or touched a toy. The recordings were rated for autism severity and dyadic engagement respectively with the Brief Observation of Social Communication Change (BOSCC, Version July 27, 2017; Grzadzinski et al., 2016) and the Joint Engagement Rating Inventory (JERI; Adamson et al., 2012: 2020) by two observers blind to the study hypotheses, group allocation and time point of the assessment. The BOSCC rates child social-communication skills (9 items), restricted and repetitive behaviors (3 items) and other cooccurring behaviors (3 items) on 6-point scales; as per the manual, averaged Total scores were obtained for two 5-minutes segments that were scored separately. The JERI characterizes child joint engagement (1 item), child availability to interact (3 items), parent support of interaction (3 items) and flow of interaction (1 item) on 7-point rating scales. The inter-rater reliability on the double-coded videos (22% of the corpus) was excellent both for the BOSCC on the Total score (intraclass correlation coefficients, ICC = 0.92) and for the JERI items (range of weighted Cohen Kappas: 0.89-1.0). The following primary outcomes were obtained in the main trial analysis: for the caregiver's outcome, Parent support of interaction measured with the JERI; for the child's outcomes, Child autism phenotypic behaviors, measured with the BOSCC, as well as Child joint engagement and Child availability to interact measured with the JERI; for the dyad's outcome, Flow of interaction measured with the JERI. For additional information on rating procedures and measures, see Salomone, Settanni, et al. (2021).

Statistical analyses

Statistical analyses were conducted as follows: first, descriptive statistics and bivariate correlations between study variables were obtained. Next, after having computed standardized effect sizes for each outcome, sequential mediation models were tested, using Hayes SPSS Process macro. These models were used to test the theoretically expected indirect effect of the intervention, as mediated by changes in parent skills supportive of the interaction (JERI Parent support of interaction). Specifically, the indirect effects of the intervention on the following outcomes were tested: BOSCC Child autism phenotypic behaviors, JERI Child joint engagement, JERI Child availability to interact, JERI Flow of interaction. For each outcome, we included in the model change scores at follow-up. Analyses were performed using change scores to measure changes in parent skills supportive of the interaction (also referred to as "parent skills"). These scores were taken at the endpoint (PSK endpoint) and at the 3-month post-intervention follow-up (PSK followup). The use of change scores is supported by literature, such as Gottman and Rushe (1993) and Rogosa and Willett (1983), which highlights their reliability and validity in capturing change over time. These scores serve as intermediate links between participation in the intervention and the study outcomes (see Figure 2). For each model, the indirect effect going only through the first mediator (change in parent skills at endpoint), the indirect effect going only through the second mediator (change in parent skills at follow-up, and the sequential indirect effect of intervention on outcome going through both mediators were calculated (Hayes, 2017; Taylor et al., 2008). Hence, for each model tested, one direct path (Intervention \rightarrow Outcome) and three indirect paths (Intevention \rightarrow PSK endpoint \rightarrow Outcome; Intervention \rightarrow PSK_followup \rightarrow Outcome; and Intervention \rightarrow PSK endpoint \rightarrow PSK followup \rightarrow Outcome) were estimated. Following Hayes et al. (2011), mediation analyses were conducted irrespective of whether a significant association, that is, a total effect, was observed between the intervention and outcome variables. In each model tested, two covariates were included: the first one, common to all models, is the measure of parent skills at baseline, the second is the outcome measure obtained at baseline. We chose to covary the pre-test outcome variable to account for potential baseline differences and to provide a clearer picture of change over time. This approach is consistent with what proposed by Castro-Schilo and Grimm (2018) and with the recommendations of researchers who advocate for controlling for baseline values to obtain a more accurate representation of change (e.g., Mara et al., 2011). Mediation analyses were carried out using Mplus ver. 8. Given the non-normality of the outcomes, we used a MLR estimator to test the model. In addition to the estimates of



FIGURE 2 Conceptual model of the effects of the intervention on study outcomes, mediated by changes in parent skills at endpoint and follow-up. Model outcomes are change scores at follow-up.

the model parameters and their respective significance levels, we also calculated indirect-to-direct and indirectto-total effects ratios for each of the study outcomes. The indirect-to-direct effects ratio shows the proportion of the total effect that is due to the indirect effect, compared to the direct effect. This ratio is useful for determining the relative importance of the indirect effect in mediating the relationship between the predictor and outcome variables. If the indirect-to-direct effects ratio is greater than 1, it suggests that the indirect effect is more important than the direct effect in explaining the relationship between the predictor and outcome variables. However, if the indirect-to-direct effects ratio is less than 1, it suggests that the direct effect is more important than the indirect effect in explaining the relationship. On the other hand, the indirect-to-total effects ratio shows the proportion of the total effect that is due to the indirect effect. compared to the total effect. This ratio is useful for determining the strength of the indirect effect relative to the overall effect.

Results

Descriptive statistics and correlations among study variables are shown in Table 1. As expected, we report strong negative correlations between child autism phenotypic behaviors and child joint engagement ratings at baseline. Change in parent skills supportive of interaction at endpoint was moderately positively correlated with change in dyad and child outcomes at follow-up, whereas there was no significant association between change in parent skills supportive of interaction at follow-up and outcomes at follow-up.

Before testing the mediation models, we calculated standardized effect sizes (Cohen's d) for each outcome. Intervention effect sizes were small for JERI scores, ranging from d = 0.21 for child availability to interact to d = 0.37 for flow of interaction. Change in child joint engagement had a value of d = 0.26. As regards autism phenotypic behaviors the effect size was absent, d = 0.03.

To assess indirect links between intervention participation and the four outcomes considered (i.e., child autism phenotypic behaviors, child joint engagement, child availability to interact, and flow of interaction) sequential mediation analyses were conducted. Results reported in Table 2 show that the effect of intervention is consistently mediated by the change in parent skills supportive of interaction. In fact, a homogeneous pattern emerges with respect to how the intervention shows to act on the considered outcomes.

In particular, it is noteworthy that the indirect effects Intervention \rightarrow PSK endpoint \rightarrow Outcome and Intervention \rightarrow PSK followup \rightarrow Outcome are always significant and in the expected direction for all outcomes considered. It is interesting to note that the indirect effect Intervention \rightarrow PSK_endpoint \rightarrow PSK_followup \rightarrow Outcome is also always significant and has the opposite effect to the indirect effects commented above. Inspection of Table 2 indicates that the sign of these indirect effects is explained by the presence of a significant negative relationship between PSK endpoint and PSK followup, that indicates that individuals who report more marked changes in parent skills supportive of interaction at baseline tend to experience a less strong change between endpoint and follow-up, and vice versa. Overall, the indirect effects for each of the outcomes considered are in line with theoretical expectations, so that participants in the intervention tend to see their parent skills supportive of interaction improve at both endpoint and follow-up more than individuals in the control group, and this improvement is reflected in an overall improvement of the outcomes of the study: the total indirect effects of the intervention are always significant and are equal to -0.12 (SE = 0.05), 0.14 (SE = 0.06), 0.12 (SE = 0.05), 0.13 (SE = 0.05) respectively for change scores in BOSCC Child autism phenotypic behaviors, JERI Child joint engagement, JERI Child availability to interact, and JERI Flow of the interaction. It is noteworthy that for all four outcomes, the direct effect is not significant. This suggests that changes in parent skills related to the intervention play a significant role in influencing the observed outcomes. Table 2 also shows the indirect-to-direct and indirect-to-total effects ratios for each outcome. Inspection of these ratios indicates that indirect effects always predominate over direct effects and account for most of the total effect of the intervention for each outcome, with minimum values of 0.59 and 0.74 respectively for JERI Flow of the

	(ISD) (SD)	1.	2.	3.	4.	5.	6.	7.	8.	9.	10.	11.
1. Age	45.21 (9.69)	I										
2. JERI Parent skills (baseline)	11.30 (4.26)	-0.10	I									
3. BOSSC Total score (baseline)	29.42 (11.26)	-0.12	-0.55**	I								
4. JERI Child Joint engagement (baseline)	3.23 (2.11)	0.09	0.65**	-0.84**	I							
5. JERI Child Availability to interact (baseline)	12.30 (4.57)	0.11	0.56**	-0.90 **	•*06.0	I						
6. JERI Flow (baseline)	3.72 (1.64)	0.04	0.67**	-0.80^{**}	0.91^{**}	0.88**	I					
7. JERI Parent skills (change T1-T2)	1.58 (4.27)	0.12	-0.41	0.23*	-0.28*	-0.31**	-0.34**	I				
8. JERI Parent skills (change T1-T3)	0.36 (4.27)	-0.12	0.14	-0.29**	0.22*	0.31**	0.28*	-0.59^{**}	I			
9. BOSSC Total score (change T1-T3)	-1.89 (6.50)	0.13	0.16	-0.20	0.14	0.15	0.12	-0.21	-0.10	Ι		
10. JERI Child Joint engagement (change T1-T3)	0.70 (1.58)	-0.08	-0.19	0.12	-0.39^{**}	-0.27*	-0.35^{**}	0.21	0.21	-0.49**	I	
11. JERI Child Availability to interact (change T1-T3)	0.54 (2.77)	-0.19	-0.05	0.04	-0.18	-0.24*	-0.17	0.34**	-0.01	-0.53**	0.55**	I
12. JERI Flow (change T1-T3)	0.58 (1.22)	-0.13	-0.22*	0.10	-0.37*	-0.26*	-0.43**	0.35**	0.14	-0.45**	0.78**	0.68**
p = 0.05; $p = 0.01$.												

-
86)
1
\geq
variables (
r study
<u>f</u> o
statistics
iptive
desci
Spearman's correlations and
TABLE 1

Outcome (change scores at follow-up)	Intervention effects	Estimate	SE	Est./ SE	<i>p</i> - Value	Indirect-to-direct effect ratio	Indirect-to-tota effect ratio
BOSCC—Total Score	Total	-0.122	0.093	-1.307	0.191		
	Direct	0.001	0.101	-0.005	0.996		
	Total indirect*	-0.121	0.052	-2.330	0.020	-121.000	0.991
	Specific indirect effects (mediated by)						
	PSK_endpoint	-0.088	0.046	-1.897	0.058		
	PSK_followup	-0.118	0.040	-2.940	0.003		
	$PSK_endpoint \rightarrow PSK_followup$	0.084	0.034	2.484	0.013		
JERI—Child Joint	Total*	0.183	0.087	2.098	0.036		
engagement	Direct	0.049	0.085	0.568	0.570		
	Total indirect*	0.135	0.061	2.224	0.026	2.755	0.738
	Specific indirect effects (mediated by)						
	PSK_endpoint	0.091	0.042	2.160	0.031		
	PSK_followup	0.153	0.055	2.772	0.006		
	PSK_endpoint→ PSK_followup	-0.109	0.037	-2.943	0.003		
JERI—Child Availability to Interact	Total	0.139	0.092	1.517	0.129		
	Direct	0.021	0.092	0.228	0.819		
	Total indirect*	0.118	0.045	2.609	0.009	5.619	0.849
	Specific indirect effects (mediated by)						
	PSK_endpoint	0.091	0.044	2.076	0.038		
	PSK_followup	0.096	0.039	2.467	0.014		
	PSK_endpoint→ PSK_followup	-0.069	0.027	-2.582	0.010		
JERI—Flow of the interaction	Total*	0.227	0.083	2.739	0.006		
	Direct	0.093	0.081	1.156	0.248		
	Total indirect*	0.133	0.052	2.556	0.011	1.430	0.586
	Specific indirect effects (mediated by)						
	PSK_endpoint	0.094	0.043	2.206	0.027		
	PSK_followup	0.136	0.047	2.918	0.004		
	PSK_endpoint→ PSK_followup	-0.097	0.036	-2.733	0.006		

SETTANNI ET AL.

-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

s and Conditions (https://onlinelibrary.wiley.com/terms

19393806, 0, Downloaded from https://onlinelibrary.wiley.com/doi/10.1002/aur. 3058 by Universita Di Torino, Wiley Online Library on [02/01/2024]. See the Terms

Note: PSK_endpoint, change in JERI Parent skills at endpoint; PSK_followup, change in JERI Parent skills at follow-up; JERI, Joint Engagement Rating Inventory; BOSCC, Brief Observation of Social Communication Change. **p* < 0.05.

interaction and JERI Child joint engagement. It should also be noted that with respect to JERI Child availability to interact and BOSCC Child autism phenotypic behaviors, the total effect of the intervention was not significant.

DISCUSSION

In this study, we tested the theory of change of, CST, a novel caregiver-mediated intervention, expecting that it

would lead to an improvement in the child and dyad outcomes through an increase in the caregiver's skills in supporting the interaction. We confirmed our a priori hypothesis showing that change in caregiver's skills supportive of interaction significantly mediated the effect of treatment on the dyad primary outcome (JERI Flow of the interaction), as well as on other child outcomes that had shown non-significant changes in the expected direction in the main trial report (Salomone, Settanni, et al., 2021): JERI child joint engagement and availability to interact, and BOSCC child autism phenotypic

behaviors. In line with theoretical expectations, we showed that parents in the CST group demonstrated greater improvements in skills in interaction with the child than those in the control group both immediately post-intervention and at the 3-months follow-up; and that this improvement was reflected in a positive change on all dyad and child outcomes at follow-up. For all outcomes, the intervention effect was significantly mediated by changes in parent skills supportive of interaction. However, it should be noted that while the indirect effect of the intervention via changes in caregiver skills at endpoint and follow-up was significant for all the studied outcomes, and the direct effect was not significant for any of them (indicating that the effect of the intervention was significantly driven by change in caregiver skills), the total effect of the intervention was significant only for flow of the dyad and child joint engagement, but not for child availability to interact and autism phenotypic behaviors. This finding may appear to contrast with our main trial analyses (Salomone, Settanni, et al., 2021), which were conducted under an ITT approach and found a significant intervention effect solely for flow of the dyad. However, the divergence in one additional outcome reaching significance is attributable to the different analytical approaches and estimators used in the two studies, each designed to answer distinct research questions.

This pattern of results distinguishes our study from the existing literature on the mechanisms underlying caregiver-mediated interventions for children with autism. Specifically, previous studies have primarily reported indirect effects on child outcomes in conjunction with significant main effects of the intervention (Brookman-Frazee et al., 2021; Carruthers et al., in press; Gulsrud et al., 2016; Pickles et al., 2015; Shih et al., 2021). This is not entirely surprising as most of these intervention models were delivered in controlled university settings (with the notable exception of Shih et al., 2021 and Brookman-Frazee et al., 2021), and all of them are more intensive than CST, a low-intensity intervention designed to be feasible even in low-resource contexts. Arguably, our mediation analysis is of particular relevance specifically because of the lack of main effects on child-level outcomes, which indicates that CST was detectably effective on the more proximal outcomes (change of caregiver's behavior, and change in the overall quality of the dyadic interaction) but not on the ultimate target, child's behavior, despite there being positive changes in the expected direction.

Formal tests of process of treatment are essential to understand treatment mechanisms and in turn to refine intervention strategies and methodologies; identifying indirect treatment effect in absence of direct or total effects is not only statistically viable and advisable (Hayes, 2017) but also clinically useful. This study suggests that the effectiveness of CST can be attributed, in part, to its focus on enhancing parent skills, which is a

key component of the intervention. This is indeed both a useful clinical information specific to CST, and a generalizable critical element of reflection for planning and design of other caregiver-mediated interventions similarly targeting early pivotal social communication behaviors. Our findings mirror indeed those of two other mechanistic studies of parent-mediated interventions, the NDBI model Improving Parents as Communication Teachers (ImPACT; Yoder et al., 2021) and the developmental intervention Adaptive Responsive Training (ART; Watson et al., 2017). Despite the trial design differences with our and previously considered mediation studies both Yoder et al. (2021) and Watson et al. (2017) are preemptive trials targeting toddlers at increased familial risk for autism, rather than diagnosed children - their pattern of results is consistent with our findings. Both studies did not find significant treatment effects on child outcomes but report moderately large intervention effects on parent skills (responsiveness in the ART trial, and overall use of a set of strategies to elicit communication and engagement in ImPACT) which in turn promoted most of the measured child outcomes (primarily communication and motor imitation). This shared pattern of findings, since it is aligned with the hypothesized theory of change, not only shows that the intervention is effective on the proximal target, caregiver's behavior, which in turn operates as catalyst of change in the ultimate target, child skills, but is also signals that there is *potential* for the intervention to produce detectable effects on these more distant targets. The question therefore becomes that of identifying and addressing the factors responsible for the attenuated treatment response.

There are indeed several reasons that may underlie a positive direct treatment effect on parent, but not child outcomes, accompanied by indirect treatment effects through the former on the latter. These are particularly worthy of examination when sample sizes are relatively small, and therefore a "full" mediation may mask other potential, unmeasured, mediators. First and foremost, it is important to acknowledge that there is substantial heterogeneity not only in the characteristics of children with autism, but also, as similarly noted by Yoder et al. (2021), in those of their parents. Several parental characteristics, such as stress, mental health concerns, education, demographic characteristics and certain cognitive profiles, are indeed likely to play a role in determining the effectiveness of caregiver-mediated interventions (Shalev et al., 2020). Gaining insight into the relationship between these important parental factors and the use of specific intervention strategies (domain-specific treatment fidelity) would help understand the complex process of acquisition of the more global "caregiver interaction skills" that ultimately impact children's outcomes. As even early intervention providers in the community tend to implement NDBI strategies unevenly (Pickard et al., 2021), it is likely also that different caregivers will find some strategies easier to incorporate in their

interaction style than others. This could be a result of the interplay among personal factors (such as parental self-efficacy; Russell & Ingersoll, 2021), perceived relevance of strategies for family functioning (Stahmer & Pellecchia, 2015), degree of complexity of implementation (e.g., use of teaching strategies is rarer prior to instruction compared to responsive strategies (Maye et al., 2020) and even cultural factors (such as respeto in Latino parenting; Adamson et al., 2021). It also remains to be explored how the rate and extent of caregiver skills acquisition is associated with relevant child characteristics, such as cognitive level, autism severity, language ability and availability to interact, which are likely candidates as moderators of treatment efficacy (Trembath et al., 2019). In light of the above-described complex dynamic of mediating components (e.g., specific intervention moderating strategies) or elements (e.g., characteristics of the parent or child) components, a second factor potentially responsible for indirect intervention effects in the absence of direct effects could be time. As Yoder et al. (2021) and Watson et al. (2017) point out, children may need more time to benefit from the mediating effects of improved parent skills, and it therefore may be necessary to implement longer followup assessments to detect smaller effects. Thirdly and finally, these findings could instead indicate the need to improve the intervention, either *qualitatively*, by refining the skills training techniques and/or modifying contents, or quantitatively, by increasing the dosage.

In sum, while our findings show that caregiver skills supportive of the interaction led the improvement in the dyad and child outcomes, learning how best to accommodate and support each caregiver's progress would improve the overall effectiveness of the intervention with the ultimate goal of ensuring progress for every dyad. This could be achieved by using the knowledge on how the ability of the parent to learn intervention strategies interacts with the child's specific needs, abilities and challenges in order to guide the design of modifications to the treatment protocol. For example, identifying which specific intervention strategies may be more difficult to learn may inform the contents and methodology of additional "booster" facilitator-led sessions or parents-only meetings, which were proposed as desirable modifications in pilot studies of CST (Salomone, Ferrante, et al., 2021; Sengupta et al., 2021). Identifying essential and desirable intervention components can also inform more costeffective use of resources, and has implications for methods of training of trainers and for planning how to best support caregivers in maintaining use of intervention strategies beyond the therapy period (Carruthers et al., 2022).

Strengths and limitations

There are several methodological strengths to the present study, which examined the treatment mechanism of one

of the first community-implemented caregiver mediated interventions evaluated through a randomized controlled design. We combined rigorous clinical characterization of participants through standardized assessments with assessment of primary outcomes that were blind rated, to a high level of reliability, at multiple timepoints, allowing a rigorous test of treatment mediation that rules out "reverse causality."

Nevertheless, the results should be interpreted in the context of some design features that may limit the interpretation of findings. The non-significance of the direct effect of the intervention on the child outcomes warrants caution in the interpretation of the findings, as outlined above. Second, even though we examined intervention effects on "flow of the dyad" that can be considered a small, and therefore potentially clinically relevant, effect as per the benchmarks generated by Chow et al. (2022), the use of primary outcome measures derived from caregiver-child interaction makes these proximal, context-bound, direct measures of social communication outcomes prone to correlated measurement error (Crank et al., 2021). Nevertheless, these measures are common in the field, and also have the strengths of objectivity through blind rating while being clinically meaningful and acceptable to caregivers. Finally, while this is among the largest RCTs conducted in a public community setting, a larger sample size would have had the necessary power to conduct subgroup analyses, for example, by autism phenotypic behaviors at baseline.

Conclusions and implications

Findings from the study demonstrate that an acceptable, feasible, open access caregiver-mediated intervention implemented in the community can lead to a sustained improvement in the quality of the caregiver-child dyadic interaction through enhancing caregiver's skills supportive of such interaction. The WHO CST has been shown to be feasible and acceptable across both low-and-middle-income countries (Montiel-Nava et al., 2022: Sengupta et al., 2021; Tekola et al., 2020) and highincome countries (Ferrante et al., 2022; Lau et al., 2022; Salomone, Ferrante, et al., 2021; Seng et al., 2022), with evidence of treatment effects on caregiver skills and overall quality of the dyadic interaction (Salomone, Settanni, et al., 2021). The current study shows that it is precisely this improvement of caregiver skills that in turn can promote social communication outcomes in the child. Further research is needed to identify and subsequently target specific caregiver characteristics and skills corresponding to intervention strategies, in order to design caregiver interventions with larger impact on children's outcomes.

AUTHOR CONTRIBUTIONS

Erica Salomone conceived and designed the study, part of the WHO field-testing initiative coordinated by

Chiara Servili, and led the research team. Helen McConachie, Katharine Suma, and Lauren B. Adamson provided training and methodological support for the coding of video-recorded data. Michele Settanni analyzed the data. Erica Salomone and Michele Settanni led the interpretation and writing of the paper with input from all authors.

ACKNOWLEDGMENTS

We wish to thank the WHO CST Team members (Felicity L. Brown, Laura Pacione, Stephanie Shire) who contributed to the development of the CST program materials and field-testing materials. We wish to thank the CST Italy Team members for institutional support and their role in data collection at each of the recruiting sites: ASL Città di Torino, Torino; ASL CN1, Mondovì and Cuneo; ASL AL, Novi Ligure; Fondazione Paideia, Torino. Finally, we are very grateful for the enormous contributions the participating families have made toward this study.

FUNDING INFORMATION

The trial was sponsored by the University of Turin, Italy. The research has received funding from the European Union's Seventh Framework program for research and innovation under the Marie Skłodowska-Curie grant agreement No. 609402-2020 researchers: Train to Move (T2M), Compagnia di San Paolo, Bando Intrecci [#2016. AAI4482.U5295], and Autism Speaks.

CONFLICT OF INTEREST STATEMENT

The authors declare no conflict of interest.

DATA AVAILABILITY STATEMENT

The data that support the findings of this study are available on request from the corresponding author. The data are not publicly available due to privacy or ethical restrictions.

ETHICS STATEMENT

This study was performed in line with the principles of the Declaration of Helsinki. Approval was granted by the Institutional Review Boards of the University of Turin (#57273) and of all National Health Service study sites: ASL Città di Torino (#0010244 and #0077761), ASL AL (#AsIAL.NPI.16.01), ASL CN1 (#180–16), ASL BI (#CE128/17).

INFORMED CONSENT STATEMENT

Informed written consent was obtained from parents/ legal guardians and professionals participating in the study.

ORCID

Lauren B. Adamson b https://orcid.org/0000-0002-6601-8874

Chiara Servili D https://orcid.org/0000-0002-1885-822X

REFERENCES

- Adamson, L. B., Bakeman, R., Deckner, D. F., & Nelson, P. B. (2012). Rating parent–child interactions: Joint engagement, communication dynamics, and shared topics in autism, Down syndrome, and typical development. *Journal of Autism and Developmental Disorders*, 42, 2622–2635.
- Adamson, L. B., Bakeman, R., & Suma, K. (2020). The joint engagement rating inventory (Technical Report 25.3). Department of Psychology, Georgia State University. https://bakeman.gsucreate.org/ DevLabTechReport25.3.pdf
- Adamson, L. B., Caughy, M. O., Bakeman, R., Rojas, R., Owen, M. T., Tamis-LeMonda, C. S., Pacheco, D., Pace, A., & Suma, K. (2021). The quality of mother-toddler communication predicts language and early literacy in Mexican-American children from low-income households. *Early Child Research Quarterly*, 56, 167–179. https://doi.org/10.1016/j.ecresq.2021.03.006
- Brookman-Frazee, L., Chlebowski, C., Villodas, M., Roesch, S., & Martinez, K. (2021). Training community therapists to deliver an individualized mental health intervention for autism spectrum disorder: Changes in caregiver outcomes and mediating role on child outcomes. *Journal of the American Academy of Child & Adolescent Psychiatry*, 60(3), 355–366. https://doi.org/10.1016/j.jaac.2020. 07.896
- Carruthers, S., Mleczko, N., Page, S., Ahuja, S., Ellis, C., Howlin, P., Leadbitter, K., Taylor, L., Slonims, V., & Charman, T. (2022). Using implementation science frameworks to explore barriers and facilitators for parents' use of therapeutic strategies following a parent-mediated autism intervention. *Autism*, 27(4), 1011–1025. https://doi.org/10.1177/13623613221125630
- Carruthers, S., Pickles, A., Charman, T., McConachie, H., Le Couteur, A., Slonims, V., Howlin, P., Collum, R., Salomone, E., Tobin, H., Gammer, I., Maxwell, J., Aldred, C., Parr, J., Leadbitter, K., & Green, J. (in press). Mediation of 6-year midchildhood follow-up outcomes after pre-school social communication (PACT) therapy for autistic children: Randomised controlled trial. *Journal of Child Psychology and Psychiatry*. https://doi.org/ 10.1111/jcpp.13798
- Castro-Schilo, L., & Grimm, K. J. (2018). Using residualized change versus difference scores for longitudinal research. *Journal of Social* and Personal Relationships, 35(1), 32–58.
- Charman, T. (2021). Editorial: One good thing (sometimes) leads to another: Demonstrating mechanistic connections between parent and child outcomes in a community implementation autism trial. *Journal of the American Academy of Child & Adolescent Psychia try*, 60(3), 338–339. https://doi.org/10.1016/j.jaac.2020.12.005
- Chow, J., Zhao, H., Sandbank, M., Bottema-Beutel, K., & Woynaroski, T. (2022). Empirically-derived effect size distributions of interventions for young children on the autism spectrum. *Journal of Clinical Child & Adolescent Psychology*, 1-13, 271–283. https://doi.org/10.1080/15374416.2021.2007485
- Crank, J. E., Sandbank, M., Dunham, K., Crowley, S., Bottema-Beutel, K., Feldman, J., & Woynaroski, T. G. (2021). Understanding the effects of naturalistic developmental behavioral interventions: A project AIM meta-analysis. *Autism Research*, 14, 817–834.
- Dawson, G., Rieder, A. D., & Johnson, M. H. (2023). A Developmental Social Neuroscience Perspective on Infant Autism Interventions. *Annual Review of Developmental Psychology*, 5, 89–113.
- Ferrante, C., Fioravanti, M., Pacione, L., Arduino, G. M., Ghersi, S., Scattoni, M. L., Team, C. I., Team, W. C., Servili, C., & Salomone, E. (2022). Supporting caregivers remotely during a pandemic: Comparison of WHO caregiver skills training delivered online versus in person in public health settings in Italy. *Journal of Autism and Developmental Disorders*. https://doi.org/10.1007/ s10803-022-05800-y
- Gottman, J. M., & Rushe, R. H. (1993). The analysis of change: Issues, fallacies, and new ideas. *Journal of Consulting and Clinical Psy*chology, 61(6), 907–910.

- Green, J., & Dunn, G. (2008). Using intervention trials in developmental psychiatry to illuminate basic science. *The British Journal of Psychiatry*, 192(5), 323–325.
- Grzadzinski, R., Carr, T., Colombi, C., McGuire, K., Dufek, S., Pickles, A., & Lord, C. (2016). Measuring changes in social communication behaviors: Preliminary development of the brief observation of social communication change (BOSCC). *Journal of Autism and Developmental Disorders*, 46(7), 2464–2479.
- Gulsrud, A. C., Hellemann, G., Shire, S., & Kasari, C. (2016). Isolating active ingredients in a parent-mediated social communication intervention for toddlers with autism spectrum disorder. *Journal of Child Psychology and Psychiatry*, 57(5), 606–613. https://doi.org/ 10.1111/jcpp.12481
- Hayes, A. F. (2009). Beyond Baron and Kenny: Statistical mediation analysis in the newmillennium. *Communication monographs*, 76(4), 408–420. https://doi.org/10.1080/03637750903310360
- Hayes, A. F. (2017). Introduction to mediation, moderation, and conditional process analysis: A regression-based approach. Guilford Publications.
- Hayes, A. F., Preacher, K. J., & Myers, T. A. (2011). Mediation and the estimation of indirect effects in political communication research. *Sourcebook for Political Communication Research*, 23(1), 434–465.
- Lau, J. S.-P., Lai, S. M.-K., Ip, F. T.-S., Wong, P. W.-C., Team, W. C., Servili, C., Salomone, E., Pacione, L., Shire, S., & Brown, F. L. (2022). Acceptability and feasibility of the World Health Organization's caregiver skills training Programme (WHO CST) delivered via eLearning, videoconferencing, and in-person hybrid modalities in Hong Kong. *Frontiers in Psychiatry*, 13. https://doi. org/10.3389/fpsyt.2022.915263
- Lord, C., Charman, T., Havdahl, A., Carbone, P., Anagnostou, E., Boyd, B., Carr, T., de Vries, P. J., Dissanayake, C., & Divan, G. (2021). The Lancet Commission on the future of care and clinical research in autism. *The Lancet*, 399(10321), 271–334.
- Lord, C., Rutter, M., DiLavore, P. C., Risi, S., Gotham, K., & Bishop, S. (2012). Autism diagnostic observation schedule (ADOS-2) (2nd ed.). Western Psychological Services.
- Mara, C. A., Cribbie, R. A., Flora, D. B., LaBrish, C., Mills, L., & Fiksenbaum, L. (2011). An improved model for evaluating change in randomized pretest, posttest, follow-up designs. *Methodology*, *8*, 97–103.
- Maye, M., Sanchez, V. E., Stone-MacDonald, A., & Carter, A. S. (2020). Early interventionists' appraisals of intervention strategies for toddlers with autism spectrum disorder and their peers in inclusive childcare classrooms. *Journal of Autism and Developmental Disorders*, 50(11), 4199–4208. https://doi.org/10.1007/s10803-020-04456-w
- Montiel-Nava, C., Tregnago, M., Marshall, J., Sohl, K., Curran, A. B., Mahurin, M., Warne-Griggs, M., & Dixon, P. (2022). Implementing the WHO caregivers skills training program with caregivers of autistic children via telehealth in rural communities. *Frontiers in Psychiatry*, 13. https://doi.org/10.3389/fpsyt.2022.909947
- Nahmias, A. S., Pellecchia, M., Stahmer, A. C., & Mandell, D. S. (2019). Effectiveness of community-based early intervention for children with autism spectrum disorder: A meta-analysis. *Journal* of Child Psychology and Psychiatry, 60(11), 1200–1209.
- Pickard, K., Mellman, H., Frost, K., Reaven, J., & Ingersoll, B. (2021). Balancing fidelity and flexibility: Usual care for young children with an increased likelihood of having autism spectrum disorder within an early intervention system. *Journal of Autism and Developmental Disorders*, 53, 656–668. https://doi.org/10.1007/s10803-021-04882-4
- Pickles, A., Harris, V., Green, J., Aldred, C., McConachie, H., Slonims, V., Le Couteur, A., Hudry, K., Charman, T., & the PACT Consortium (2015). Treatment mechanism in the MRC preschool autism communication trial: Implications for study design and parent-focussed therapy for children. *Journal of Child*

Psychology and Psychiatry, 56(2), 162–170. https://doi.org/10. 1111/jcpp.12291

- Rogosa, D. R., & Willett, J. B. (1983). Demonstrating the reliability of the difference score in the measurement of change. *Journal of Educational Measurement*, 20, 335–343.
- Russell, K. M., & Ingersoll, B. (2021). Factors related to parental therapeutic self-efficacy in a parent-mediated intervention for children with autism spectrum disorder: A mixed methods study. *Autism*, 25(4), 971–981. https://doi.org/10.1177/1362361320974233
- Salomone, E., Ferrante, C., Salandin, A., Ferrara, F., Torchio, E., Foletti, G., Ghersi, S., Team, C. I., Team, W. C., Pacione, L., & Servili, C. (2021). Acceptability and feasibility of the World Health Organization's caregiver skills training implemented in the Italian National Health System. *Autism*, 26(4), 859–874. https://doi. org/10.1177/13623613211035228
- Salomone, E., Pacione, L., Shire, S. Y., Brown, F., Reichow, B., & Servili, C. (2019). Development of the WHO caregiver skills training programme for developmental disorders or delays. *Frontiers in Psychiatry*, 10, 769.
- Salomone, E., Settanni, M., McConachie, H., Suma, K., Ferrara, F., Foletti, G., Salandin, A., Brown, F. L., Pacione, L., Shire, S., Servili, C., Adamson, L. B., & Team, W. C. (2021). Pilot randomized controlled trial of the WHO caregiver skills training in public health services in Italy. *Journal of Autism and Developmental Disorders*, 52, 4286–4300. https://doi.org/10.1007/s10803-021-05297-x
- Schreibman, L., Dawson, G., Stahmer, A. C., Landa, R., Rogers, S. J., McGee, G. G., Kasari, C., Ingersoll, B., Kaiser, A. P., & Bruinsma, Y. (2015). Naturalistic developmental behavioral interventions: Empirically validated treatments for autism spectrum disorder. *Journal of Autism and Developmental Disorders*, 45(8), 2411–2428.
- Seng, G.-J., Chiu, Y.-N., Tsai, W.-C., Lin, H.-Y., Li, S.-C., Hsiao, M.-N., Liu, T.-J., Chen, H.-M., Shih, A., Chang, Y.-C., Team, W. H. O. C. S. T., & Soong, W.-T. (2022). Promotion and implementation effectiveness of World Health Organization's caregiver skills training program in Taiwan. *Frontiers in Psychiatry*, 13. https://doi.org/10.3389/fpsyt.2022.904380
- Sengupta, K., Shah, H., Ghosh, S., Sanghvi, D., Mahadik, S., Dani, A., Deshmukh, O., Pacione, L., Dixon, P., & Salomone, E. (2021). World Health Organisation-caregiver skills training (WHO-CST) program: Feasibility of delivery by non-specialist providers in realworld urban settings in India. *Journal of Autism and Developmental Disorders*, 53, 1444–1461.
- Shalev, R. A., Lavine, C., & Di Martino, A. (2020). A systematic review of the role of parent characteristics in parent-mediated interventions for children with autism spectrum disorder. *Journal of Devel*opmental and Physical Disabilities, 32(1), 1–21.
- Shih, W., Shire, S., Chang, Y.-C., & Kasari, C. (2021). Joint engagement is a potential mechanism leading to increased initiations of joint attention and downstream effects on language: JASPER early intervention for children with ASD. *Journal of Child Psychology* and Psychiatry, 62(10), 1228–1235. https://doi.org/10.1111/jcpp. 13405
- Stahmer, A. C., & Pellecchia, M. (2015). Moving towards a more ecologically valid model of parent-implemented interventions in autism. *Autism*, 19(3), 259–261. https://doi.org/10.1177/ 1362361314566739
- Taylor, A. B., MacKinnon, D. P., & Tein, J.-Y. (2008). Tests of the three-path mediated effect. Organizational Research Methods, 11(2), 241–269.
- Tekola, B., Girma, F., Kinfe, M., Abdurahman, R., Tesfaye, M., Yenus, Z., Team, W. C., Salomone, E., Pacione, L., & Fekadu, A. (2020). Adapting and pre-testing the World Health Organization's caregiver skills training programme for autism and other developmental disorders in a very low-resource setting: Findings from Ethiopia. *Autism*, 24, 51–63.
- Trembath, D., Gurm, M., Scheerer, N. E., Trevisan, D. A., Paynter, J., Bohadana, G., Roberts, J., & Iarocci, G. (2019). Systematic review

of factors that may influence the outcomes and generalizability of parent-mediated interventions for young children with autism spectrum disorder. *Autism Research*, *12*(9), 1304–1321. https://doi.org/10.1002/aur.2168

- Watson, L. R., Crais, E. R., Baranek, G. T., Turner-Brown, L., Sideris, J., Wakeford, L., Kinard, J., Reznick, J. S., Martin, K. L., & Nowell, S. W. (2017). Parent-mediated intervention for one-year-olds screened as at-risk for autism spectrum disorder: A randomized controlled trial. *Journal of Autism and Developmental Disorders*, 47(11), 3520–3540. https://doi.org/10. 1007/s10803-017-3268-0
- WHO. (2022). Caregiver skills training for families of children with developmental delays or disabilities: Introduction. World Health Organization. https://iris.who.int/handle/10665/353579
- Williams, N. J., & Beidas, R. S. (2019). Annual research review: The state of implementation science in child psychology and psychiatry: A review and suggestions to advance the field. *Journal of Child Psychology and Psychiatry*, 60(4), 430–450.

Yoder, P. J., Stone, W. L., & Edmunds, S. R. (2021). Parent utilization of ImPACT intervention strategies is a mediator of proximal then distal social communication outcomes in younger siblings of children with ASD. *Autism*, 25(1), 44–57. https://doi.org/10.1177/ 1362361320946883

How to cite this article: Settanni, M., Suma, K., Adamson, L. B., McConachie, H., Servili, C., & Salomone, E. (2023). Treatment mechanism of the WHO caregiver skills training intervention for autism delivered in community settings. *Autism Research*, 1–13. https://doi.org/10.1002/aur.3058